11 WORKING PAPERS 2022

THE SOLVENCY AND FUNDING COST NEXUS – THE ROLE OF MARKET STIGMA FOR BUFFER USABILITY

Helena Carvalho | Lucas Avezum Fátima Silva



WORKING PAPERS 2022

THE SOLVENCY AND FUNDING COST NEXUS – THE ROLE OF MARKET STIGMA FOR BUFFER USABILITY

Helena Carvalho | Lucas Avezum Fátima Silva

JULY 2022

The analyses, opinions and findings of these papers represent the views of the authors, they are not necessarily those of the Banco de Portugal or the Eurosystem.

> Please address correspondence to Banco de Portugal Rua do Comércio 148, 1100-150 Lisboa, Portugal Tel.: +351 213 130 000, email: info@bportugal.pt



Lisboa, 2022 • www.bportugal.pt

 Working Papers
 | Lisboa 2022 • Banco de Portugal Rua do Comércio 148 | 1100-150 Lisboa • www.bportugal.pt •

 Edition Banco de Portugal • ISBN (online) 978-989-678-836-0 • ISSN (online) 2182-0422

Helena Carvalho Banco de Portugal Lucas Avezum Banco de Portugal

Fátima Silva Banco de Portugal

July 2022

Abstract

In this paper, we investigate the relationship between the banks' solvency ratio and their funding costs using a proprietary dataset from Banco de Portugal for 21 Portuguese banks from 2006 to 2020. In light of the discussion on impediments to capital buffer usability by banks, we focus on the importance of market discipline to this relationship. Our results suggest that the relationship between solvency and funding costs is negative and state-dependent, i.e. market participants become more sensitive to changes in solvency during economic downturns. The relationship is stronger for market-based financing sources in comparison to deposits. Finally, we use a breakpoint analysis and find that investors are more likely to penalize the same absolute deterioration in solvency levels when banks are already in a fragile position. Our findings support the hypothesis that fear of market stigma may make banks reticent to use buffers in times of stress.

JEL: G00, G21, G28 Keywords: solvency, funding costs, capital requirements, market stigma.

Acknowledgements: We are grateful to Bruno Nascimento for his assistance in accessing the data. This paper benefited greatly from Vítor Oliveira, Daniel Abreu, Ana Pereira, Ana Cristina Leal, Inês Drumond, Pedro Duarte Neves, and Luísa Farinha helpful comments and suggestions. The views expressed in this article are those of the authors and do not necessarily reflect the views of Banco de Portugal or the Eurosystem. Any errors and mistakes are ours.

E-mail: hcarvalho@bportugal.pt; lavezum@bportugal.pt; mfsilva@bportugal.pt

1. Introduction

The Great Financial Crisis exposed vulnerabilities in the quality and quantity of banks' capital. It was the catalyst for increasing regulatory capital requirements, including the introduction of macroprudential buffers that can be used during economic downturns to incentivize banks to continue providing credit to the economy instead of engaging in excessive de-leveraging or de-risking behaviors. However, market pressure to maintain or even increase capital ratios can constrain banks in using their buffers during economic downturns. Hence, concerns with higher funding costs can be one of the reasons why capital ratios in Europe did not decrease meaning that banks did not dip into their buffers during the onset of the COVID-19 crisis despite a large release of regulatory capital buffer (Couaillier *et al.* 2021).

Understanding the relationship between banks' solvency and funding costs is crucial for prudential policy. It has been argued that increasing capital requirements is costly for banks, as they are forced to change their funding composition towards more expensive sources of funding. However, higher solvency levels also make the bank safer leading to lower risk premiums being demanded. The effect of solvency on funding costs depends on both the effect it has on the rates of individual funding sources and on the funding composition. A rise in funding costs erodes banks' profitability which might impair banks' ability to generate the required capital levels internally, an essential tool for withstanding unexpected shocks. In order to preserve capital, banks might respond by curtailing credit provision to the real economy either by passing the costs to customers (price channel) and/or by reducing the supply of credit (quantity channel). A higher cost of credit could impact negatively on overall economic activity as the number of borrowers that become unable to repay their loans may rise too. Moreover, excessive de-leveraging or de-risking strategies could have severe adverse effects on the real economy.

In this paper, we investigate the relationship between the solvency ratio and funding costs using a proprietary dataset from Banco de Portugal for 21 Portuguese banks from 2006 to 2020. Our measure for funding cost is the ratio between the interest paid to the stock of interest-bearing liabilities and it does not include equity financing. The empirical literature studying the determinants of funding costs points towards a negative relationship, i.e. better solvency, asset quality, profitability, or liquidity are cost reducing. In light of the discussion on the effectiveness of the current regulatory capital framework, we focus in particular on the importance of market discipline to this relationship. First, we expect the relationship to be state-dependent as market participants become more risk-averse during economic downturns and may require banks to maintain higher capital ratios to reduce default risk. The long time span of our dataset and the fact that it is on a quarterly basis means that we can investigate if the relationship is stronger during crisis periods. Second, our detailed balance sheet dataset allows us to calculate the funding costs for different financing sources. We expect a stronger relationship between solvency and market-based financing sources in comparison to deposits.

Finally, investors are more likely to penalize the same absolute deterioration in solvency levels when banks are already in a fragile position. To test this hypothesis, we use breakpoint analysis to study if the relationship is non-linear. Importantly, to account for the possible endogeneity between solvency and funding costs, besides using standard panel OLS estimation and fixed effects, we use an instrumental variable strategy. The use of multiple specifications in each section of the paper works as a robustness check.

We find that solvency, measured as the ratio of tier 1 capital to risk-weighted assets, is negatively associated with funding costs. A reduction in the *Solvency Ratio* of 100 bps is associated with an increase in *Funding Cost* of between 1 and 10 bps, depending on the estimator and specification used. We also find that the relationship is state-dependent. During a crisis, the effect of a 100bps increase in the *Solvency Ratio* is 15bps more negative compared to normal periods. The impact of changes in solvency on the cost of each individual funding component (deposits, interbank, claims on central banks, or debt) is also more pronounced during crisis periods. We observe a stronger relationship for funding sources more sensitive to market pressure, such as interbank lending and debt securities than for deposits. The paper also provides evidence that the marginal benefit on funding costs of increasing capital ratio is not constant for all levels of capital. Above a certain level of solvency - depending on the specification, between 11% and 16% - the estimated coefficient of the *Solvency Ratio* on *Funding Cost* is substantially smaller or even positive, depending on the estimator and specification used.

We contribute to the literature that studies the interaction between funding costs and solvency. Several papers find a negative association between solvency and funding costs (Aldasoro et al. 2022; Arnould et al. 2021; Aymanns et al. 2016; Babihuga et al. 2014; Elyasiani and Keegan 2017; Gambacorta and Shin 2016; Hasan et al. 2016; Schmitz et al. 2017). Most research looks at the determinants of market-based debt funding cost rates. Market-based rates have the advantage of capturing best the marginal effect on funding costs from changes in bank fundamentals; investors monitor banking indicators and quickly adjust their investment positions accordingly. However, banks have several sources of funding with prices that react differently to changes in fundamentals. If the cost of one funding source changes, banks can optimize their funding composition. As such, studies focusing on the cost of an individual source of funding may over- or underestimate the effect of solvency. Arnould et al. (2021) find that senior bond yields are the most sensitive to changes in solvency, and that deposit rates are the least sensitive. Aymanns et al. (2016) also find that solvency is negatively related to both the average and wholesale funding cost; the coefficient is larger in magnitude for wholesale funding, pointing to the higher credit risk sensitivity of wholesale investors. Conversely, our approach allows us to decompose the overall funding cost measure according to the funding source. However, ideally, we would like to relate the solvency level to the effective interest rate of each funding source.

Market participants' sensitivity to risk is dependent upon the phase of the economic cycle, as shown by Elyasiani and Keegan (2017) that find that the

importance of market discipline is stronger during periods of crisis and post-crisis. In Babihuga *et al.* (2014), the interaction of the subprime crisis dummy with the capital variable suggests that changes in bank capital matter more for funding costs during and after the crisis. In our analysis, we find similar results for the funding cost and contribute to the literature by showing that the state dependency also holds for the relationship between solvency and the cost of each funding source. The existence of a state-dependent effect has implications for our understanding of the degree of complementarity between regulatory action and market discipline. The release of buffers may not be effective if market forces work in the opposite direction, by tightening scrutiny and punishing banks that make use of the reduced requirements to decrease their level of solvency.

The marginal effect on funding costs of increasing solvency is not linear. The effect of a unit increase in solvency on funding costs is larger when solvency levels are low, as observed by Aymanns *et al.* (2016) and Schmitz *et al.* (2017). Arnould *et al.* (2021) also noticed the existence of threshold beyond which the effect of an increase in solvency on the chosen measure of funding cost changes. Our findings show that, only below a certain level of solvency, an increase in solvency decreases the funding costs, increasing the banks' profitability and their ability to internally generate short-term capital.

The paper proceed as follows: Section 2 presents the dataset used in our analysis, related descriptive statistics, and stylized facts. Section 3 presents the main results and Section 4 includes robustness and extensions. Finally, Section 5 concludes.

2. Data

The paper considers quarterly time series data for 21 Portuguese banking groups over the period from the first quarter of 2006 to the last quarter of 2020. We restrict the sample to financial institutions with their main activity in Portugal that accept deposits. The sample corresponds to the Portuguese banking system excluding branches of credit institutions having their head office in other European Union Member-States, to ensure that they have agency over their funding decisions. Bank-specific information is taken from 'Séries Longas para o Setor Bancário Português', a panel dataset produced by Banco de Portugal (Couchinho *et al.* 2019). Furthermore, we used internal supervisory data sources to extend the dataset with more granular information on balance sheet and P&L items. We complement the bank-specific information with macro-financial variables from the ECB Statistical Data Warehouse (SDW). All the variables were winsorized at 1% level to minimize the influence of outliers.

Table 1 provides descriptive statistics of our main explanatory and dependent variables. There is significant variation in funding cost rates, as well as in bank-specific characteristics. This reflects the long time span of the sample as well as the heterogeneous features of each financial institution and provides a diverse setting

to test our main hypothesis of a negative relationship between solvency and the funding costs faced by the institution. Detailed variable definitions and sources are provided in Table A.1 of the Appendix.

Variable	Num. Obs.	Mean	Std. Dev.	Pctl. 25	Pctl. 75
Funding Cost	1155	0.018	0.015	0.008	0.025
Deposits Funding Cost	1159	0.014	0.010	0.005	0.021
Central Bank Funding Cost	699	0.011	0.021	0.001	0.011
Interbank Funding Cost	1114	0.016	0.022	0.001	0.021
Debt Funding Cost	697	0.030	0.019	0.016	0.043
Solvency Ratio	1147	0.203	0.123	0.112	0.269
Provisions to Assets	1164	0.003	0.005	0.000	0.005
Loan Loss Reserves Ratio	1162	0.057	0.050	0.025	0.077
Average Risk Weight (RW)	1147	0.581	0.161	0.473	0.663
Liquidity Ratio	1164	0.038	0.043	0.014	0.043
Return on Equity (ROE)	1164	0.024	0.079	0.006	0.058
Total assets (€10 ⁹)	1164	19.001	30.459	0.264	31.055
Size	1164	7.560	2.526	5.575	10.343
EONIA	1185	0.006	0.014	-0.004	0.008
Dow Jones Euro Stoxx 50 Index	1185	3.255	0.582	2.718	3.733
V2TX	1185	0.228	0.080	0.168	0.263
Deposits Annualized Agreed Rate	1168	0.017	0.014	0.004	0.027

Table 1. Summary Statistics

This table presents summary statistics for our sample, which comprises 21 Portuguese financial institutions. The data span the period from the first quarter of 2006 to the last quarter of 2020. Variable definitions and sources are provided in table A.1 of the appendix.

The main dependent variable, *Funding Cost*, is annualized interest and other similar charges divided by total interest-bearing liabilities. Implicit measures of funding costs from accounting data might not be as marginally responsive as market-based metrics, such as Credit Default Swap (CDS) spreads or bond yields, but provide a more accurate view of banks' actual funding costs, given its funding mix composition. Furthermore, our choice allows us to compute the cost of each source of non-equity funding and understand the effect of funding mix adjustments on the overall funding cost.

For this purpose, we compute measures of the funding costs for the main liabilities components using the ratio of interest paid for that component to the respective stock of liabilities: *Deposit Funding Cost* for deposits from households, non-financial corporations, government, and other financial corporations, *Central Bank Funding Cost* for liquidity provided by central banks, *Interbank Funding Cost* for deposits from credit institutions, and *Debt Funding Cost* for funding from issuing bonds or other debt instruments. We expect different sensitivities of each funding cost measure, with deposit funding being the least sensitive and interbank/debt funding being the most sensitive.

Figures 1 and 2 present the evolution of the non-equity funding costs and the aggregate funding mix composition. Over the sample period, funding costs

substantially decreased from an average funding cost of 3% in the first quarter of 2006 to an average of 0.5% in the last quarter of 2020. It is possible to distinguish four stages in the evolution of the funding costs of Portuguese banks. From 2006 to the GFC of 2008-09, funding costs were rising and displayed significant heterogeneity across the banks in the sample. Debt and deposit funding dominated aggregate bank funding expenses, and interbank funding also played a significant role, even though relatively smaller than the previous sources. From the GFC to the onset of the Portuguese sovereign debt crisis in 2010 there was a marked decrease in funding costs, and Portuguese banks reduced their reliance on interbank funding. Instead, there was an increase in central bank funding expenses while the relative importance of debt and deposit funding was kept. From 2010 to the end of 2012, funding costs started to increase again. It was during this period that Portuguese banks most relied on central bank funding and shifted their funding model towards more traditional sources of funding, like deposits. From 2012 onwards, the trend has been for a continuous decrease in funding costs, in line with the decrease in interest rates. Overall, the stress from the GFC exacerbated by the tumultuous years of the sovereign debt crisis led to a change in the funding structure of Portuguese banks. They reduced their reliance on interbank and debt funding sources and turned to more traditional sources of funding like deposits. The ratio of customer loans to deposits decreased from an average of 142% in the first quarter of 2006 to 79% in the last quarter of 2020. Recently, Portuguese banks have maintained their reliance on deposit funding but have also started to rely more on debt funding, which could be linked to new regulatory requirements like the MREL (Minimum Requirement for own funds and Eligible Liabilities).

Overall, the developments of the last fifteen years are the product of a series of structural developments. The financial crisis changed the funding model of Portuguese banks. We see an effort to hedge against abrupt changes in wholesale market rates and conditions that may take place during a crisis, reflected in a higher share of deposit funding. The advantages of flexible sources of funding like debt and interbank lending, however, are its downfall during periods of crisis when these sources are less reliable in terms of market access and conditions. Indeed, over-reliance on short-term wholesale funding was proven detrimental to financial stability (Oura *et al.* 2013). The low-interest-rate environment from an accommodative monetary policy stance is the main driver in the continuous decrease in funding costs. Notwithstanding the generalized structural development, there is significant heterogeneity between the financial institutions in the sample. We investigate the role of bank-specific characteristics in explaining this variation and also compare their relative importance against broader macro-financial factors.

To test our measure of implicit funding cost, we have also collected observed interest rates that are individually agreed between the monetary institution and the customer for each new deposit (Deposits Annualized Agreed Rate), per bank and per quarter. The correlation between the observed deposit rates for new operations and our implicit *Deposit Funding Cost* is 0.82.





Funding Cost is computed as annualized interest and other similar charges divided by total interest bearing liabilities. The graph shows a boxplot for the distribution of Funding cost for the banks in the sample in each quarter. The lower and upper hinges correspond to the 25th and 75th percentiles, the lower/upper whisker extends 1.5 * inter-quartile range from the hinge. Data beyond the whisker are plotted individually.



Figure 2: Evolution of the composition of funding costs, in terms of interest expenses and correspondent liability stock.

The panel on the left shows the composition of interest expenses by source of funding (in $M \in$). The panel on the right shows the composition of funding liabilities (in $M \in$). The non-equity funding sources considered were Deposits from households, non-financial corporations, government, other financial corporations, deposits from Central Banks, Interbank deposits from credit institutions, Debt securities.



Figure 3: Evolution of the Solvency Ratio.

Solvency Ratio is computed as Tier 1 Capital divided by total risk-weighted assets. The graph shows a boxplot for the distribution of *Solvency Ratio* for the banks in the sample in each quarter. The lower and upper hinges correspond to the 25th and 75th percentiles, the lower/upper whisker extends 1.5 * inter-quartile range from the hinge. Data beyond the whisker are plotted individually.

The main explanatory variable, Solvency Ratio, is the Tier 1 Own Funds ratio, which is a measure of high-quality regulatory capital in relation to risk-weighted assets. Tier 1 capital is going concern capital comprising Common Equity Tier 1 and Additional Tier 1 capital - it includes common shares and stock surplus, retained earnings, other comprehensive income, capital instruments meeting the criteria for Additional Tier 1 and related surplus, qualifying minority interests, and regulatory adjustments. We selected the Tier 1 ratio as our measure of solvency because it only considers capital instruments, share premiums, and retained profits that are available to absorb losses. A possible alternative would have been to use the total capital ratio as the measure for solvency. In our view, doing so would be inappropriate because the measure includes subordinated loans that from an accounting point of view are classified as debt obligations and thus simultaneously included in our funding cost measure. Over the sample period, the banks strengthened their Solvency Ratio from an average of 19% (median 13%) in the first quarter of 2006 to an average of 24% (median 18%) in the last quarter of 2020. The average increase in the Solvency Ratio is also the result of the change in the regulatory landscape that raised minimum capital requirements.

We control for size, liquidity, asset quality, and profitability. The size of banking groups is measured by the logarithm of total assets. We expect to observe a negative relationship between bank size and funding costs which might be related to monitoring costs, meaning that smaller institutions may face difficulties in accessing funding from the market. In turn, more liquidity, captured by the cash ratio (cash and deposits in central banks divided by total assets), enhances the ability to meet

short-term liabilities and thus should be perceived favorably by the markets. Asset quality is measured by the loan loss reserve ratio. A lower share of impaired loans in total customer loans should be priced positively by the market. Finally, we use the return on equity as a measure of profitability. The expected relationship between profitability and funding costs is ambiguous because lower profitability levels may call into question the sustainability of the institution, while extremely high levels may also be perceived as indicating excessive risk-taking.

Some specifications also use the Euro Overnight Index Average (EONIA). We prefer EONIA because it is anchored in actual financial transactions to obtain the weighted average of unsecured overnight interbank lending, whereas Euribor relies on quotes issued by the panel banks which has raised concerns about its reliability. In robustness checks, not presented for brevity sake, we re-estimated all specifications using Euribor but the results do not change materially. Additionally, we include the Dow Jones Euro Stoxx 50 Price Index to capture the market sentiment. For market volatility, we use V2TX, the VSTOXX index.

3. The relationship between solvency and funding costs

In this section, we analyze: (i) the sign of the solvency and funding costs relationship; (ii) state-dependency, to assess whether the economic and financial cycle position affects the relationship causing it to be different in normal times versus crisis times; (iii) the sensitivity of different funding sources to changes in solvency; (iv) the existence of non-linearity in the relationship with respect to the level of solvency.

As a whole, the different specifications address the shortcomings identified by the literature, including possible endogeneity due to second order effects between solvency and funding costs. We examine the relationship between the *Solvency Ratio* and the *Funding Cost*, using three specifications:

$$FC_{it} = \beta_0 + \beta_1 solv_{it} + \beta X_{it} + \gamma Y_t + \varepsilon_{it} \tag{1}$$

$$FC_{it} = \beta_0 + \beta_1 solv_{it} + \beta X_{it} + b_i + f_t + \varepsilon_{it}$$
⁽²⁾

$$solv_{it} = \beta_0 + \beta_1 RW_{it} + \beta X_{it} + b_i + f_t + \varepsilon_{it}$$
(3.1)

$$FC_{it} = \beta_0 + \beta_1 \widehat{solv}_{it} + \beta X_{it} + b_i + f_t + \varepsilon_{it}$$
(3.2)

where FC_{it} is the Funding Cost for each bank in each quarter. The independent variable of interest is $solv_{it}$, the Solvency Ratio. Specification (1) corresponds to a linear panel data model with vector X_{it} for time-varying bank-specific variables for Size, the Loan Loss Reserves Ratio, the Liquidity Ratio, and the Return on Equity Ratio and vector Y_t for the macro-financial controls EONIA, Dow Jones, and V2TX for volatility. Specification (2) includes fixed effects b_i and f_t for bank and quarter, respectively. The simultaneous inclusion of time and bank fixed effects controls for any remaining variability not being captured by the model. Specification (3.1) and (3.2) decompose the instrumental variable estimation into its two stages, where (3.1) obtains predicted values for the possibly endogenous regressor $solv_{it}$ using risk-weights (RW_{it}) as the instrument and (3.2) regresses the dependent variable on the predicted values of $solv_{it}$ and other control variables. There might be a significant nexus of second order effects not captured by the specifications used so far. To account for endogeneity, we use instrumental variables. Standard errors are corrected for heteroscedasticity in most specifications.

Table 2 results show a negative relationship between the *Solvency Ratio* and *Funding Cost*. Column (1) shows the results of specification (1) including banklevel control variables. We obtain a statistically significant coefficient of -0.042 suggesting a negative relationship between solvency and funding costs. Column (2) includes global market variables: the Euro Overnight Index Average rate (EONIA), the quarterly Dow Jones Euro Stoxx Price Index, and a volatility index (V2TX). The use of macro-financial controls provides insights on the relative importance of bank-specific characteristics vis-à-vis global market influences. In this specification, controlling for overall market conditions reduces the magnitude of the coefficient on solvency but it remains significant and negative. The sensitivity of overall funding cost to changes in EONIA behaves as expected. Better overall market performance, as measured by the Dow Jones index, also has the effect of reducing the funding costs of banks. Higher market wide volatility, captured by the V2TX, decreases banks' funding costs. In times of market stress, funding providers may display a 'flight to safety' behavior and lean towards instruments with less uncertainty.

To account for omitted variable bias, specifications in columns (3), (4), and (5) introduce bank and year-quarter fixed effects. The simultaneous inclusion of time and bank fixed effects causes the coefficient on solvency to lose significance, visible in column (5). However, as noted by on Banking Supervision (2015) and Schmitz *et al.* (2017) the relationship between solvency and funding costs may not be unilateral and there might be a significant nexus of second-order effects not captured by fixed effects. Indeed, high funding costs erode profitability with an effect on the solvency indicator.¹

To account for the possible source of endogeneity, we use instrumental variable estimators. An instrument is considered valid if it fulfills two conditions: (i)

^{1.} A variable addition regression-based test provides evidence of endogeneity of *Solvency Ratio* supporting the use of 2SLS.

exogeneity and (ii) relevance. We tested two instruments, Provisions to Assets in columns (6) and (7) similarly to Schmitz et al. (2017) and the Risk Weight (RW) in columns (8) and (9). While the rationale for the exogeneity of *Provisions* to Assets seems valid - provisions directly reduce solvency but will only affect funding costs through its effect on the solvency ratio and, in turn, provisions are not directly affected by financing costs - the instrument does not fulfill the relevance condition. The instrument relevance test indicates a weak instrument.² Conversely, the F-statistic in column (8) indicates that the *Risk Weight* has explanatory power in regard to the Solvency Ratio. Moreover, the Risk Weight fulfills the exogeneity condition because it directly affects the Solvency Ratio but only affects funding costs through its effect on solvency and, in turn, we don't expect financing costs to affect risk-weights contemporaneously. Implicit is also the assumption that the average risk weight of the banks' is not widely monitored by market participants. a reasonable assumption considering the time span of the sample and the nature of the banks in it. In the last specification using the IV estimator (9), to a 100 bps increase in the Solvency Ratio reduces Funding Cost by approximately 9.9 bps.

Overall, Table 2 suggests that when faced with a negative solvency shock, debtholders' (and depositors') reaction to the implicit increase in default probability is to demand higher yields. This result is in line with the estimates obtained in the literature, which suggest that a 100 bps increase in solvency is associated with a decrease in funding costs of between 2 bps (Aymanns *et al.* 2016) and 23 bps (Aldasoro *et al.* 2022).³

^{2.} We consider a case with one endogenous variable and one instrument k. When k=1, $F^E ff = F^R$ can be compared to Stock and Yogo (2005) critical values, as suggested in Andrews *et al.* (2019).

^{3.} The estimates have a wide range and use different proxies for solvency and for funding costs. Notwithstanding the limitations of comparing estimators from very distinctive analysis, the existing literature finds that, all other factors constant, banks with a higher solvency level benefit from lower funding costs.

Dependent Variable:	Poo	led OLS		Ξ			ì	<	
Funding Cost	(1)	(2)	(3)	(4)	(5)	(6) ¹ 1SLS	(7) 2SLS	(8)1 SLS	
Colorado Datio	-0.042***	-0.016***	-0.014***	-0.010*	-0.004		0.121		
	(0.004)	(0.002)	(0.002)	(0.004)	(0.004)		(0.275)		
0:12	0.001**	0.002***	0.002***	0.006***	0.008***	-0.062***	0.015	-0.081***	
JILE	(0.000)	(0.000)	(0.000)	(0.001)	(0.001)	(0.010)	(0.017)	(0.010)	
Dan Done Donewing Datio	-0.030***	-0.004	-0.011*	-0.007	-0.018***	0.018	-0.021*	-0.129*	
LUAN LUSS Reserves Raulu	(0.007)	(0.004)	(0.005)	(0.005)	(0.005)	(0.050)	(0.010)	(0.053)	
in idity Datio	-0.094***	-0.005	0.004	-0.008	0.008	-0.077	0.017	-0.126*	
	(0.008)	(0.005)	(0.005)	(0.007)	(0.008)	(0.055)	(0.023)	(0.050)	
ROF	0.011 +	-0.007+	-0.012**	-0.007	-0.012***	0.070*	-0.023	0.122***	
	(0.006)	(0.004)	(0.004)	(0.004)	(0.003)	(0.028)	(0.024)	(0.025)	
EONIA		0.713*** (0.028)		0.730*** (0.028)					
		-0.010***		-0.010***					
		(0.001)		(0.001)					
V2TX		(0.005)		(0.004)					
Drowinings to Anoth						-0.295			
Provisions to Assets						-0.426			
BW								-0.188***	
								-0.021	
Bank fixed effects	No	No	No	Yes	Yes	Yes	Yes	Yes	
Year-Quarter fixed effects	No	No	Yes	No	Yes	Yes	Yes	Yes	
Num. Obs.	1138	1138	1138	1138	1138	1145	1138	1145	
R ²	0.269	0.654	0.696	0.739	0.787	0.849	0.626	0.87	
R ² Adj.	0.266	0.651	0.678	0.733	0.77	0.837	0.597	0.86	
F-Test statistic for						0.4813		77.925	
instrument significance									

12

Table 2. Negative relationship between solvency and funding costs

of 2006 to the last quarter of 2020. Variable definitions and sources are provided in table A.1 of the appendix. Column (1) to (2) provide estimates for model specification (1). Column (1) is a baseline estimate for *Funding Cost* dependent on the *Solvency Ratio* and other bank-characteristics. Column (2) From the F-Test statistic for instrument significance in column (8) we conclude that the instrument is relevant. Column (9) is the second stage least square estimation. + p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001 | Standard errors in brackets | ¹The dependent variable is the *Solvency Ratio*. Column (3) includes year-quarter fixed effects, column (4) includes bank fixed effects, and column (5) includes both. Columns (6) and (7) provide estimates for model specification (3.1) and (3.2) using Provisions to Assets as the instrument. Column (6) presents the first stage least square estimation. From the use this instrument are not valid. Column (8) and (9) provide estimates for model specification (3.1) and (3.2) using Risk Weight (RW) as the instrument. F-Test statistic for instrument significance we conclude that the instrument does not fulfill the relevance condition. As such, the estimates in column (7) that adds controls for macro-financial variables that vary each quarter but not for each bank. Columns (3) to (5) provide estimates for model specification (2). This table reports regressions of Funding Cost on the Solvency Ratio, controlling for bank-specific characteristics. The time period is from the first quarter

3.1. The state-dependent relationship between solvency and funding costs

In this section, we test the hypothesis of the relationship between the *Solvency Ratio* and the *Funding Cost* being different during moments of crisis relative to normal times. We extend the prior analysis of Table 2 by interacting a crisis indicator variable with solvency using the following specifications:

$$FC_{it} = \beta_0 + \beta_1 solv_{it} + \beta_2 crisis_t + \beta_3 solv_{it} * crisis_t + \beta X_{it} + \gamma Y_t + \varepsilon_{it}$$
(4)

$$FC_{it} = \beta_0 + \beta_1 solv_{it} + \beta_2 crisis_t + \beta_3 solv_{it} * crisis_t + \beta X_{it} + b_i + f_t + \varepsilon_{it}$$
(5)

$$solv_{it} = \beta_0 + \beta_1 RW_{it} + \beta_2 crisis_t + \beta_3 RW_{it} * crisis_t + \beta X_{it} + b_i + f_t + \varepsilon_{it}$$
(6.1)

$$FC_{it} = \beta_0 + \beta_1 \widehat{solv}_{it} + \beta_2 crisis_t + \beta_3 \widehat{solv}_{it} * crisis_t + \beta X_{it} + b_i + f_t + \varepsilon_{it}$$
(6.2)

where the variable $crisis_t$ takes the value of one if the year-quarter falls into a crisis period, and zero otherwise. The definition of the crisis period uses the reference business cycle chronology for the Portuguese economy based on the monthly coincident indicator disclosed by Banco de Portugal and developed by Rua (2017). The monthly coincident indicator incorporates information from real GDP and other relevant economic variables that capture the overall state of the economy. From 2006 to 2020, there were two recessions in Portugal. The first from the last quarter of 2007 to the second quarter of 2009, associated with the Great Financial Crisis. The second recession occurred from the third quarter of 2010 to the second quarter of 2013, corresponding to the period of the sovereign debt crisis in Portugal.

Table 3 shows that the association between the *Solvency Ratio* and *Funding Cost* of banks is stronger during periods of crisis. Column (3), our preferred specification using *RW* as the instrument, confirms that during periods of crisis, funding costs are more sensitive to changes in solvency. Indeed, during a crisis, the effect of a 100bps increase in the *Solvency Ratio* is 15bps more negative during a crisis compared to normal periods. Additionally, higher levels of solvency also help reduce the financing costs even in periods without crisis, albeit to a less extent. Our results are consistent with the hypothesis that market participants' sensitivity to risk is dependent upon the phase of the economic cycle. For instance, Elyasiani and Keegan (2017) find that the sensitivity of the yield of bonds issued by US

systemically important banks to time-varying bank-specific variables is lower in the pre-crisis period compared with the crisis and post-crisis period.

The negative link between solvency and funding costs differs in normal times versus crisis times. Understanding the variation in the relationship between funding costs and solvency throughout the cycle has an important implication to understand the degree of complementarity between regulatory action and market discipline. One of the main novelties of the macroprudential framework created in the aftermath of the financial crisis was the introduction of a countercyclical buffer that is released during downturns to incentivize banks to dip into their capital buffers and continue providing credit to the economy instead of engaging in excessive de-leveraging or de-risking behaviors. This channel of transmission may be impaired if market forces work in the opposite direction, by tightening scrutiny and punishing banks that make use of the reduced requirements to decrease their level of solvency. If banks are aware that investors in general attribute higher importance to high levels of solvency during periods of stress, they act rationally by being more reluctant to tap into their capital buffers.

Dependent Variable:	Pooled OLS	FE	IV
Funding Cost	(1)	(2)	(3)
Solvency Ratio	-0.008**	0.004	-0.039*
	(0.003)	(0.004)	(0.020)
Crisis	0.004**		
	(0.001)		
Solvency Ratio $ imes$ Crisis	-0.024***	-0.022***	-0.151***
	(0.005)	(0.004)	(0.045)
Size	0.002***	0.007***	-0.003
	(0.000)	(0.001)	(0.003)
Loan Loss Reserves Ratio	0.003	-0.011*	0.031*
	(0.004)	(0.005)	(0.015)
Liquidity Ratio	0.001	0.013 +	0.035**
	(0.005)	(0.007)	(0.013)
ROE	-0.006+	-0.010**	0.008
	(0.004)	(0.003)	(0.007)
EONIA	0.735***		
	(0.031)		
Dow Jones	-0.010***		
	(0.001)		
V2TX	-0.019***		
	(0.005)		
Bank fixed effects	No	Yes	Yes
Year-Quarter fixed effects	No	Yes	Yes
Num.Obs.	1138	1138	1138
R ²	0.662	0.793	0.504
R ² Adj.	0.659	0.777	0.465

Table 3. Relationship between solvency and funding costs in crisis times

This table reports regressions of *Funding Cost* on the *Solvency Ratio*, with an interaction term for periods of crisis and controlling for bank-specific characteristics. The time period is from the first quarter of 2006 to the last quarter of 2020. Variable definitions and sources are provided in table A.1 of the appendix. Column (1) provides estimates for model specification (4) and shows that the effect of solvency on funding costs has a larger magnitude during periods of crisis. Column (2) provides estimates for model specification (5). The result of state-dependency is robust to the inclusion of fixed effects. Column (3) provides estimates for the IV model specification (6.1) and (6.2). + p <0.1, * p <0.05, ** p <0.01, *** p <0.001 | Standard errors in brackets.

3.2. Sensitivity of funding sources to changes in solvency

In this section, we look at the relationship between the *Solvency Ratio* and the funding cost of each funding source. We estimate the state-dependency econometric specifications (4), (5), (6.1.), and (6.2.) but change the dependent variable to capture the individual funding cost sources. The overall funding cost measure FC_{it} is decomposed according to the funding source (deposits, interbank, central banks, or debt) using the ratio of interest paid for that component to the respective stock of liabilities.

Table 4 presents the results of each specification for each of the funding source components. The first three columns have *Deposit Funding Cost* as the dependent variable. The results show that during a crisis, solvency and funding cost from

depositors have a negative relationship. Relative to the other sources of funding, the coefficient for the relationship between the *Solvency Ratio* and *Deposit Funding Cost* is the smallest, both in normal and in crisis times. Economic rationale suggests that deposits should indeed be the least sensitive and interbank/debt funding the most sensitive to changes in solvency. The deposit guarantee scheme works as an implicit subsidy (Bahaj and Malherbe 2020). Its existence creates a funding advantage, depositors' funds are insured in the event of bank default. Trust from depositors makes it easier to capture deposits and these are a cheaper source of financing. The safety net enjoyed by depositors reduces their incentive to monitor the health of the financial institution where they chose to deposit funds (Boot *et al.* 1993; Demirgüç-Kunt and Huizinga 2004).

Columns (4)-(6) have the *Central Bank Funding Cost* as the dependent variable. We use the information for deposits from central banks to compute the ratio between annualized interest and the stock of deposits. The coefficients for normal times are not statistically significant . The lack of sensitivity of this funding rate to solvency reflects that central banks' financing operations make liquidity available in exchange for adequate collateral with a haircut according to the type of collateral. Conversely, during a crisis, the coefficient is statistically significant in column (6). The negative association during a crisis is compatible with the euro area monetary policy framework to lend freely against good collateral in moments of crisis but only to solvent banks.

The columns (7)-(9) and (10)-(12) examine the relationship between the *Solvency Ratio* and *Interbank* and *Debt Funding Cost*, respectively. For *Interbank Funding Cost*, we use data for deposits from credit institutions to compute the ratio between annualized interest and the stock of deposits. For *Debt Funding Cost*, we use the ratio of annualized interest pertaining to debt securities and the stock of these instruments. Columns (7)-(9) show that *Interbank Funding Cost* is highly sensitive to changes in *Solvency Ratio* both in normal and even more so during crisis times. Column (8), in particular, shows that during a crisis the effect of an increase of 100bps in the *Solvency Ratio* reduces the *Interbank Funding Cost* by approximately 2.5bps more in comparison to calmer periods. The magnitude of this difference is substantial since during normal times the effect of a 100bps increase in the *Solvency Ratio* on the *Interbank Funding Cost* is already a reduction of 2.9bps. Column (12) paints a similar picture, the sensitivity of *Debt Funding Cost* during periods of crisis is of a similar magnitude albeit with lower statistical significance.

As expected, the magnitude of the coefficient is highest in the case of *Interbank* and *Debt Funding Cost* – especially so during moments of crisis. Our results lend empirical weight to the hypothesis put forward by King (2008) that the effect of market discipline on the interbank market would constitute an upper bound among other liabilities. Banks benefit from a comparative advantage given their expertise in the activity and are better equipped to monitor other credit institutions. Investors and other banks alike can be considered the most sophisticated market participants and carefully monitor banking fundamentals.

Regarding the statistical significance of the coefficients, our approach has a meaningful drawback. Our measures of *Interbank* and *Debt Funding Cost* are implicit average rates from accounting data that are not quite as marginally responsive. The rigidity of our selected proxies is a limitation for the study of this relationship. This limitation is of particular importance in the case of *Debt Funding Cost*, especially because Portuguese banks have greatly reduced their reliance on debt funding over the sample period. The reduction in the stock of debt funding means that our proxy measure of funding cost is less reflective of the marginal cost of this source of funding, which might explain our somewhat weak result.

		Deposits			Central Bank			Interbank			Debt	
	Pooled OLS	FE	<	Pooled OLS	FE	<	Pooled OLS	FE	₹	Pooled OLS	FE	₹
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Solvency Ratio	-0.017***	-0.001	0.022*	-0.003	0.031	0.208	-0.019**	-0.029*	-0.159**	-0.078***	-0.010	0.047
	0.002+	()	()	0.000	(0.000)	()	0.010**	(,)	()	0.002	()	(2222)
Crisis	(0.001)	-	-	(0.003)	-		(0.003)	-		(0.003)		
Solvency Ratio \times Crisis	-0.007*	-0.009***	-0.100***	-0.023	-0.056***	-0.365*	-0.033***	-0.025**	-0.121	-0.006	-0.020	-0.143*
	(0.003) 0.000***	(0.002)	_0.026)	(0.015) 0.000	_0 017)	_0.149)	(0.008) 0.001*	_0.009)	_0.075)	(0.020) 0.001**	(0.014)	_0.060)
Size	(0.000)	(0.001)	(0.002)	(0.000)	(0.004)	(0.012)	(0.000)	(0.003)	(0.008)	(0.000)	(0.003)	(0.008)
Loan Loss Reserves Ratio	0.011**	-0.006	(n nna)	0.006	0.017	0.024	(0.029+	-0.007 (0.018)	0.020	-0.009	-0.029	-0.039
in idit. Datio	-0.006	0.022***	0.042***	0.139****	0.164***	0.193***	-0.031*	0.026	0.011	0.022	-0.025	-0.018
	(0.004) _0 011***	_0.005)	0.008)	0.032)	_0.038)	_0.043)	(0.013)	(0.017) 0.008	(0.026) 0.032*	0.021) _0 057***	(0.032)	(0.032)
ROE	(0.003)	(0.002)	(0.004)	(0.006)	(0.007)	(0.015)	(0.009)	(0.008)	(0.015)	(0.011)	(0.010)	(0.011)
FONIA	0.463***			0.995***			0.437***			0.254***		
	(0.017)			(0.149)			(0.068)			(0.068)		
Dow Jones	-0.010***			-0.006*			0.000			-0.001		
VJTY	-0.023***			-0.039**			0.017			-0.021+		
	(0.003)			(0.014)			(0.014)			(0.012)		
Bank fixed effects	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes
Year-quarter fixed effects	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes
Num.Obs.	1141	1141	1141	694	694	694	1098	1098	1098	685	685	685
, R ^r	0.642	0.871	0.635	0.276	0.446	0.257	0.172	0.335	0.194	0.205	0.55	0.512
R ² Adj.	0.639	0.861	0.606	0.266	0.376	0.162	0.164	0.281	0.127	0.193	0.492	0.449

Table 4. Sensitivity of different funding cost sources to changes in the level of solvency

in table A.1 of the appendix. The first main column titled 'Deposits' looks at deposits from households, non-financial corporations, government, and other financial corporations and uses the ratio between annualized interest expenses and the stock of deposits as the dependent variable. The second main column pertaining to debt securities and the stock of these instruments. The specifications in column (1) to (12) refer to models analogous to those in Table 3. + p credit institutions and the correspondent stock of deposits. The fourth main column titled 'Debt' uses as dependent variable the ratio of annualized interest stock of deposits. The third main column titled 'Interbank' uses as dependent variable the ratio between annualized interest expenses from deposits from titled 'Central Bank' uses as dependent variable the ratio between annualized interest expenses from deposits from Central Banks and the correspondent This table reports regressions of each individual funding cost source rate on the Solvency Ratio, with an interaction term for periods of crisis and controlling for bank-specific characteristics. The time period is from the first quarter of 2006 to the last quarter of 2020. Variable definitions and sources are provided <0.1, * p <0.05, ** p <0.01, *** p <0.001 | Standard errors in brackets.

3.3. Non-linearity depending on the level of solvency

In this section, we test the hypothesis of non-linearity in the relationship between the *Solvency Ratio* and the *Funding Cost*. We use two different types of models. First, we extend our baseline specifications (1) and (2) by employing a segmented or broken-line model as in (7) and (8):

$$FC_{it} = \beta_0 + \beta_1 solv_{it} + \beta_2 (solv_{it} - \varphi)_+ + \beta X_{it} + \gamma Y_t + \varepsilon_{it}$$
(7)

$$FC_{it} = \beta_0 + \beta_1 solv_{it} + \beta_2 (solv_{it} - \varphi)_+ + \beta X_{it} + b_i + f_t + \varepsilon_{it}$$
(8)

where $(solv_{it} - \varphi)_+ = (solv_{it} - \varphi)I(solv_{it} > \varphi)$ and $I(\cdot)$ is the indicator function equal to one when the statement is true. The parameter φ is the breakpoint, in our case it is the level of solvency upon which the relationship between solvency and funding costs changes. Thus, in this model we estimate a coefficient for the effect of solvency on funding costs for values of solvency below the estimated breakpoint and another for values above the breakpoint. The advantage of this methodology is both its interpretability and its ability to provide the breakpoint estimate. We follow Muggeo (2003) approach that reduces the problem to an iterative fitting of linear models.⁴

Table 5 shows that there exists a significant difference-in-slope regarding the effect on *Funding Cost* of a change in the *Solvency Ratio* between banks that in a given quarter have solvency above the breakpoint and those with solvency below. We performed Davies' test for the specifications in all columns to conclude that there exists a breakpoint, which is evidence of non-linearity.⁵ Column (1) presents the results of estimating model (7). The breakpoint is estimated at a *Solvency Ratio* of 12.5%. The funding costs of banks with a *Solvency Ratio* below 12.5% are highly sensitive to changes in solvency – a 100bps increase in the *Solvency Ratio* above 12.5%, the effect has a smaller magnitude - a 100bps increase in the *Solvency Ratio* above 12.5%, the effect has a smaller magnitude - a 100bps. Next, in column (2), we add bank and year-quarter fixed effects to account for possible omitted variable bias. The estimation yields a breakpoint estimate of 15.7%. Again we see the same pattern of higher sensitivity of *Funding Cost* to changes in the *Solvency Ratio* cost by *Solvency Ratio* cost by *Solvency Ratio* possible omitted variable bias.

^{4.} For more information about the method see Muggeo (2003) for a discussion of the estimation method, Muggeo (2008) for an illustration, Muggeo (2016) for an approach to hypothesis testing, and Muggeo (2017) for a discussion of interval estimation for the breakpoint.

^{5.} Muggeo proposes employing the Davies (1987) test for testing the hypothesis of the differencein-slopes parameter being equal to zero. More information about the test in Muggeo (2008).

for banks with lower solvency. Banks that have solvency above the breakpoint face a small but statistically significant increase in *Funding Cost* suggesting that the investors no longer place as much importance on changes in solvency when evaluating banks with an already high solvency position.

	Piecewise I	regression
	Pooled OLS	FE
	(1)	(2)
Solvency Ratio below Breakpoint	-0.220***	-0.160***
	(0.028)	(0.018)
Solvency Ratio above Breakpoint	-0.009*	0.008 +
	(0.003)	(0.004)
Size	0.001***	0.004***
	(0.000)	(0.001)
Loan Loss Reserves Ratio	0.003	-0.011**
	(0.005)	(0.005)
Liquidity Ratio	-0.001	0.018**
	(0.006)	(0.007)
ROE	-0.006	-0.008**
	(0.003)	(0.003)
EONIA	0.619***	
	(0.023)	
Dow Jones	-0.009***	
	(0.001)	
V2TX	-0.019***	
	(0.005)	
Bank fixed effects	No	Yes
Year-quarter fixed effects	No	Yes
Num.Obs.	1138	1138
R ²	0.6784	0.8063
R ² Adj.	0.6756	0.7907
	0.125	0.157
Breakpoint Estimation	(0.005)	(0.006)
Davies Test for a change in the slope	Ò O Í	Ò Ó

Table 5. Non-linear relationship between solvency and funding cost

This table reports non-linear regressions of the *Funding Cost* on the *Solvency Ratio*, controlling for bank-specific characteristics. The time period is from the first quarter of 2006 to the last quarter of 2020. Variable definitions and sources are provided in table A.1 of the appendix. In the estimation we used piecewise regression to estimate the *Solvency Ratio* at which the breakpoint occurred and estimate slopes before and after the breakpoint. Segmented regression was implemented using the R package 'segmented' developed by Vito M. R. Muggeo. The methodology tests for the presence of an inflection point and estimates its location. Column (1) presents the results of estimating specification (7) and column (2) the results of estimating model (8). The first row includes the coefficient for the slope before the breakpoint and the second row provides the coefficient for the slope after the breakpoint. The estimated breakpoint is also provided in the table. + p <0.1, * p <0.05, ** p <0.01, *** p <0.001 | Standard errors in brackets

The second step of the analysis is to test the existence of non-linearity in model (3), in which we used 2SLS. At the moment, to our knowledge, there is not a proven robust way to apply segmented models in an instrumental variable setting. Instead, we use a parametric change point analysis. Scheines et al. (2001) propose a method for creating a piecewise linear IV-estimator ($PL-IV^*$) that consists of identifying the breakpoint location in the explanatory variable and then applying the IV estimator separately in each region. To identify the change-point in solvency, we use the Schwarz Information Criteria (SIC) as in Chen and Gupta (2012) informational approach. We compute the SIC for the null hypothesis H_0 of the model being linear and for the alternative hypothesis Ha containing the set of non-linear models with a change point k at positions 2, 3, to n-2. The decision rule for selecting the model is to select the model with a change at k for which $SIC(k) = min(SIC(H_a))$ if $min(SIC(H_a)) < SIC(H_0)$. After identifying the change point, we divided the sample into two regions - one above the change point and another below the change point. A downside of the PL-IV* methodology is that it may underestimate the standard error of the other parameters since it does not capture the uncertainty in the change point estimation.

Table 6 presents the results of applying the PL-IV*. First, this method also favors the existence of a non-linear relationship with an estimated solvency change point of 11%. We divide the sample by the change point and separately use Risk Weight as instruments to 2SLS estimate each region. The *Funding Cost* of banks with a *Solvency Ratio* below 11% are highly sensitive to changes in solvency – a 100bps increase in the *Solvency Ratio* reduces *Funding Cost* by approximately 58.5 bps. For banks with a *Solvency Ratio* above 11%, the effect has a smaller magnitude - a 100bps increase in the *Solvency Ratio* reduces *Funding Cost* by approximately 5.4 bps. The results are consistent with Aymanns *et al.* (2016) who also find that the effect of an additional increase in solvency on wholesale funding costs is more than twice as large for poorly capitalized banks relative to well-capitalized banks.

	PL-	·IV*
	Solvency Ratio	Solvency Ratio
	below Breakpoint	above Breakpoint
Solvency Ratio	-0.585+	-0.054***
	(0.351)	(0.011)
Size	0.013	0.000
	(0.009)	(0.002)
Loan Loss Reserves Ratio	0.035	0.011 +
	(0.058)	(0.006)
Liquidity Ratio	0.007	0.014+
	(0.045)	(0.007)
ROE	-0.017	-0.009*
	(0.013)	(0.005)
Bank fixed effects	Yes	Yes
Year-quarter fixed effects	Yes	Yes
Num.Obs.	259	879
R ²	0.824	0.710
R ² Adj.	0.768	0.680
Breakpoint Estimation	0.1	110

Table 6. Non-linear relationship between solvency and funding cost using PL-IV*

This table reports regressions of the *Funding Cost* on the *Solvency Ratio. Solvency Ratio* is instrumented using Risk Weights. The time period is from the first quarter of 2006 to the last quarter of 2020. Variable definitions and sources are provided in table A.1 of the appendix. At the moment, to our knowledge, there isn't a proven robust way to apply segmented models in an instrumental variable setting. Instead, we use parametric change point analysis, as proposed in Scheines et al. (2001). + p <0.1, * p <0.05, ** p <0.01, *** p <0.001 | Standard errors in brackets

The existence of capital requirements helps explain the non-linearity in the relationship. Banks seek to maintain a positive distance between their capital ratio and the overall capital requirement, i.e., a voluntary buffer, to avoid distribution restrictions and market stigma. Our estimated breakpoints, between 11% and 16%, are around the average tier 1 capital requirement ratio in our sample (12.2%), which supports the relevance of market stigma in explaining why capital buffers were not used during crisis times. The findings suggest that investors penalize more the same absolute deterioration in solvency levels when the bank's capital ratio is below or close to the overall capital requirement.

4. Robustness and Extensions

4.1. Deposits Annualized Agreed Rate

In our main analysis, we use the ratio between the interest paid to the stock of interest-bearing liabilities as the measure of funding cost. This approach has the advantage of allowing us to also calculate measures of the funding cost rates of

the different funding sources. However, ideally, we would like to relate the solvency level to the effective interest rate of each funding source.

As a robustness test of our main dependent variables, we use an alternative measure of funding costs that reflects the interest rate banks pay. On a monthly basis, Banco de Portugal computes the average, weighted by their amounts, of interest rates that are individually agreed between the credit institution and the customer for each new deposit. While this measure should better capture the sensitivity of funding costs to solvency, it has the disadvantage of being only available for deposit funding. Nevertheless, because deposits are the largest funding source for Portuguese banks, the results of this test are meaningful for the overall *Funding Cost*. After averaging the monthly data to obtain year-quarter observations, we re-estimate the state-dependency specifications using *Deposits Annualized Agreed Rate* as the dependent variable.

Table 7 provides the results. In column (3) we observe that in normal times, depositors are not sensitive to changes in the level of solvency; as previously stated. During a crisis, the effect of a 100bps increase in the *Solvency Ratio* reduces *Deposits Annualized Agreed Rate* by approximately 5.2bps more in comparison to normal periods. This effect is equal in sign but smaller in magnitude than that registered for our computed implicit measure of *Deposits Funding Cost*.

		Deposits	
	(Deposits Ar	nnualized Agi	reed Rate)
	Pooled OLS	FE	IV
	(1)	(2)	(3)
Solvency Ratio	-0.017***	-0.007*	0.001
	(0.002)	(0.003)	(0.010)
Crisis	0.009***		
	(0.001)		
Solvency Ratio $ imes$ Crisis	-0.016***	-0.019***	-0.052*
	(0.004)	(0.003)	(0.021)
Size	-0.001***	-0.002**	-0.004**
	(0.000)	(0.001)	(0.001)
Loan Loss Reserves Ratio	-0.001	-0.004	0.006
	(0.004)	(0.004)	(0.008)
Liquidity Ratio	0.010*	0.020***	0.028***
	(0.004)	(0.006)	(0.008)
ROE	-0.001	0.003	0.006*
	(0.003)	(0.002)	(0.003)
EONIA	0.741***		
	(0.019)		
Dow Jones	-0.009***		
	(0.001)		
V2TX	-0.032***		
	(0.003)		
Bank fixed effects	No	Yes	Yes
Year-quarter fixed effects	No	Yes	Yes
Num.Obs.	1136	1136	1136
R ²	0.793	0.897	0.881
R ² Adj.	0.791	0.889	0.872

Table 7. Sensitivity of Deposits Annualized Agreed Rate to changes in solvency

This table reports regressions of the Deposits Annualized Agreed Rate on the *Solvency Ratio*, with an interaction term for periods of crisis and controlling for bank-specific characteristics. In column (3) the *Solvency Ratio* is instrumented using *RW*. The time period is from the first quarter of 2006 to the last quarter of 2020. Variable definitions and sources are provided in table A.1 of the appendix. + p <0.1, * p <0.05, ** p <0.01, *** p <0.001 | Standard errors in brackets.

4.2. The interaction with non-conventional monetary policy

The period studied coincided with an environment of economic recovery and decreasing interest rates driven by increasingly loose monetary policy stances. Monetary policy is a global factor that we control for through the use of yearquarter fixed effects in our main specifications. Even so, we are interested in better understanding how the non-conventional monetary policy decisions of the period, like large scale asset purchase programs, can affect the relationship between solvency and funding cost. To capture all dimensions of monetary policy, we use the shadow rate developed by Wu and Xia (2020) that reflects conventional and non-conventional monetary policy tools when an economy reaches the effective lower bound.

	Average Fi	Inding Cost	Deposit Funding Cost	Deposits Annualized Agreed Rate
	FE	IV	IV	IV
	(1)	(2)	(3)	(4)
Solvency Ratio	-0.014***	-0.082***	-0.013+	-0.020*
	(0.004)	(0.014)	(0.007)	(0.009)
Solvency Ratio $ imes$ Shadow Rate	-0.570***	-1.821***	-0.460**	-0.403*
	(0.046)	(0.378)	(0.169)	(0.179)
Size	0.005***	-0.002	-0.003*	-0.003*
	(0.001)	(0.002)	(0.001)	(0.001)
Loan Loss Reserves Ratio	-0.009+	0.013 +	-0.001	-0.003
	(0.005)	(0.008)	(0.004)	(0.005)
Liquidity Ratio	0.026***	0.063***	0.035***	0.029***
	(0.007)	(0.015)	(0.008)	(0.008)
ROE	-0.011***	-0.005	-0.003	0.003
	(0.003)	(0.004)	(0.002)	(0.002)
Num.Obs.	1138	1138	1141	1136
R ²	0.809	0.68	0.849	0.899
R ² Adj.	0.794	0.655	0.837	0.891

Table 8. Interaction with monetary policy

This table reports the results of introducing the *Shadow Rate* in the specification as an interaction term with *Solvency Ratio* and controlling for bank-specific characteristics. In column (1) and (2) the dependent variable is the *Funding Cost*. In column (3) the dependent variable in the *Deposit Funding Cost* and in column (4) the dependent variable is the *Deposits Annualized Agreed Rate*. The time period is from the first quarter of 2006 to the last quarter of 2020. Variable definitions and sources are provided in table A.1 of the appendix. + p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001 | Standard errors in brackets.

Table 8 presents the results. As the *Shadow Rate* decreases, i.e. monetary policy loosens, the *Funding Cost* becomes less sensitive to changes in solvency. The loss of sensitivity of deposit funding costs to solvency as monetary policy loosens is visible in column (3) and (4) both for the *Deposit Funding Cost* and for the *Deposits Annualized Agreed Rate*. Proximity to the Zero Lower Bound may create an accumulation of deposit rates offered at a near zero level if banks are not allowed to pass on that negative rate to clients. The loss in sensitivity to solvency is stronger for overall funding costs than for deposit funding costs, as visible by comparing column (1) and (2) with columns (3) and (4). Additional to deposits, *Funding Cost* encompasses interbank and debt funding sources. In this case, the loss of funding cost's sensitivity to fundamentals like solvency is compounded by the documented search-for-yield behavior.

5. Conclusion

Promoting a banking sector with an adequate level of solvency is key for macroprudential policy. Adequacy presupposes that the banking system is able

to increase the solvency level during the upside of the business/financial cycle and, in turn, use the capital to absorb the effects of a shock during a stress event. The signal and magnitude of the relationship between solvency and funding costs have received increased attention as a potential impediment to capital buffer usability.

Using data for the Portuguese banking system, we find that the relationship between funding costs and solvency is negative and stronger during a crisis than in normal times. Market funding sources, notably interbank lending and debt securities, are more sensitive to changes in solvency than deposits. This finding has an important implication to assess the degree of complementarity between regulatory action and market discipline. Evidence of heightened perceived risk during crisis – especially in market funding sources – helps explain why, despite the large capital requirements release during the Covid-19 crisis, banks were reluctant to use the capital buffers. The effectiveness of a release of a capital buffer may be impaired if market forces work in the opposite direction, by tightening scrutiny and punishing banks that make use of the reduced requirements to reduce their level of solvency.

We also find evidence of non-linearity of funding costs with respect to the level of solvency. Indeed, only below a certain level of solvency, an increase in solvency decreases the funding costs, increasing the banks' profitability and their ability to internally generate short-term capital. This result validates the current minimum capital requirements regulatory design that works as a floor that protects banks from going below the optimal level that would leave them exposed to harsh investors' oversight and large penalization in the case of further decreases.

The existence of state dependency and non-linearity in the negative relationship between funding costs and solvency supports the relevance of market stigma in explaining why capital buffers were not used during crisis times.

Variable	Definition and source
Funding Cost rate (FC)	Annualized interest and other similar charges divided by total interest bearing liabilities. Each bank records the amount of interest paid up to each quarter. To annualize the data, we divide by the number of quarters that have passed and then multiply by four to obtain the yearly equivalent. Séries Longas para o Setor Bancário Português, Banco de Portugal
Deposits Funding Cost rate (Dep)	Annualized interest and other similar charges from deposits from households, non-financial corporations, government, and other financial corporations divided by total deposits from customers. Internal supervisory data, Banco de Portugal; Séries Longas para o Setor Bancário Português, Banco de Portugal

6. Appendix

Funding Cost rate (CB) resources divided by total deposits from Central Banks. Internal supervisory data, Banco de Portugal Interbank Funding Annualized interest and other similar charges from deposits from credit institutions divided by total deposits from credit institutions. Internal supervisory data, Banco de Portugal. Debt Funding Annualized interest and other similar charges from debt securities cost rate (IB) Debt Funding Annualized interest and other similar charges from debt securities cost rate (Debt) Solvency Ratio Tier 1 Capital divided by total risk-weighted assets. Tier 1 capital is the sum of common shares and stock surplus, retained earnings, other comprehensive income, qualifying minority interest, capital instruments meeting the criteria from AT1 and related surplus, additional qualifying minority interest and regulatory adjustments. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Provisions to Provisions and impairments (net of reversals) divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Loan Loss Impairments and value adjustments in customer loans divided by gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de Portugal Risk-weights Average risk-weight of exposure amount. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Risk-weights <th>Central Bank</th> <th>Annualized interest and other similar charges from Central Bank</th>	Central Bank	Annualized interest and other similar charges from Central Bank
 (CB) supervisory data, Bánco de Portugal; Séries Longas para o Setor Bancário Português, Banco de Portugal Interbank Funding Annualized interest and other similar charges from deposits from credit institutions divided by total deposits from credit institutions. Internal supervisory data, Banco de Portugal; Séries Longas para o Setor Bancário Português, Banco de Portugal Debt Funding Annualized interest and other similar charges from debt securities cost rate (Debt) Post Funding Annualized interest and other similar charges from debt securities cost rate (Debt) Solvency Ratio Tier 1 Capital divided by total risk-weighted assets. Tier 1 capital is the sum of common shares and stock surplus, retained earnings, other comprehensive income, qualifying minority interest, capital instruments meeting the criteria from AT1 and related surplus, additional qualifying minority interest and regulatory adjustments. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Provisions to Provisions and impairments (net of reversals) divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Loan Loss Impairments and value adjustments in customer loans divided by gross carrying amount of customer loans. Séries Longas para o Setor Bancário Portuguê, Banco de Portugal Risk-weights Average risk-weight of exposure amount. Risk-weighted assets are all assets. Séries Longas para o Setor Bancário Portugual. Rik-weight Cash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Portugal Liquidity Ratio Cash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Portugal Riturn on Equity Net profit or loss divided by total equity. Séries Longas para o	Funding Cost rate	resources divided by total deposits from Central Banks. Internal
Bancário Português, Banco de Portugal Interbank Funding Annualized interest and other similar charges from deposits from credit institutions divided by total deposits from credit institutions. Internal supervisory data, Banco de Portugal; Séries Longas para o Setor Bancário Português, Banco de Portugal Debt Funding Annualized interest and other similar charges from debt securities issued and other financial liabilities divided by total liabilities represented by debt titles. Internal supervisory data, Banco de Portugal Solvency Ratio Tier 1 Capital divided by total risk-weighted assets. Tier 1 capital is the sum of common shares and stock surplus, retained earnings, other comprehensive income, qualifying minority interest, capital instruments meeting the criteria from AT1 and related surplus, additional qualifying minority interest and regulatory adjustments. Risk-weighted assets are all assets held by the bank weighted by total assets. Series Longas para o Setor Bancário Português, Banco de Portugal Provisions to Provisions and impairments (net of reversals) divided by total assets. Series Longas para o Setor Bancário Português, Banco de Portugal Loan Loss Impairments and value adjustments in customer loans divided by total assets. Series Longas para o Setor Bancário Português, Banco de Portugal Liquidity Ratio Average risk-weight of exposure amount. Risk-weighted assets are all assets. Heid by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Liquidity Ratio Cash and cash balances/loans to central banks divided by total assets. Séries Longas para	(CB)	supervisory data. Banco de Portugal: Séries Longas para o Setor
Interbank Funding Cost rate (IB) Annualized interest and other similar charges from deposits from credit institutions divided by total deposits from credit institutions. Internal supervisory data, Banco de Portugal: Setor Bancário Português, Banco de Portugal Debt Funding Annualized interest and other similar charges from debt securities issued and other financial liabilities divided by total liabilities represented by debt titles. Internal supervisory data, Banco de Portugal. Solvency Ratio Tier 1 Capital divided by total risk-weighted assets. Tier 1 capital is the sum of common shares and stock surplus, retained earnings, other comprehensive income, qualifying minority interest, capital instruments meeting the criteria from AT1 and related surplus, additional qualifying minority interest and regulatory adjustments. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Provisions to Assets Provisions and impairments (net of reversals) divided by total Assets. Series Longas para o Setor Bancário Português, Banco de Portugal Loan Losa Impairments and value adjustments in customer loans divided by Reserves gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de Portugal Risk-weight Average risk-weight of exposure amount. Risk-weighted assets are all (RW) assets Logas para o Setor Bancário Português, Banco de Portugal Liquidity Ratio Cash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco d		Bancário Português, Banco de Portugal
Cost rate (IB) credit institutions divided by total deposits from credit institutions. Internal supervisory data, Banco de Portugal Debt Funding Cost rate (Debt) Annualized interest and other similar charges from debt securities issued and other financial liabilities divided by total liabilities represented by debt titles. Internal supervisory data, Banco de Portugal; Séries Longas para o Setor Bancário Português, Banco de Portugal Solvency Ratio Tier 1 Capital divided by total risk-weighted assets. Tier 1 capital is the sum of common shares and stock surplus, retained earnings, other comprehensive income, qualifying minority interest, capital instruments meeting the criteria from AT1 and related surplus, additional qualifying minority interest and regulatory adjustments. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Provisions to Provisions and impairments (net of reversals) divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Loan Loss Impairments and value adjustments in customer loans divided by gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de Portugal Risk-weights Average risk-weight of exposure amount. Risk-weighted assets are all (RW) assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Liquidity Ratio Cash and cash balances/loans to central banks divided	Interbank Funding	Annualized interest and other similar charges from deposits from
Internal supervisory data, Banco de Portugal; Séries Longas para o Setor Bancário Português, Banco de Portugal Debt Funding Cost rate (Debt) Annualized interest and other similar charges from debt securities represented by debt titles. Internal supervisory data, Banco de Portugal; Séries Longas para o Setor Bancário Português, Banco de Portugal Solvency Ratio Tier 1 Capital divided by total risk-weighted assets. Tier 1 capital is the sum of common shares and stock surplus, retained earnings, other comprehensive income, qualifying minority interest, capital instruments meeting the criteria from AT1 and related surplus, additional qualifying minority interest and regulatory adjustments. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Provisions to Provisions and impairments (net of reversals) divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Loan Loss Impairments and value adjustments in customer loans divided by gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de Portugal Risk-weights Average risk-weight of exposure amount. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Liquidity Ratio Cash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Size Logarithm of total assets. Séries Longas par	Cost rate (IB)	credit institutions divided by total deposits from credit institutions.
Setor Banćário Português, Banco de PortugalDebtFundingCost rate (Debt)Annualized interest and other similar charges from debt securitiesissued and other financial liabilities divided by total liabilitiesrepresented by debt titles. Internal supervisory data, Banco dePortugal;Séries Longas para o Setor Bancário Português, Banco dePortugalTier 1 Capital divided by total risk-weighted assets. Tier 1 capitalis the sum of common shares and stock surplus, retained earnings, other comprehensive income, qualifying minority interest, capital instruments meeting the criteria from AT1 and related surplus, additional qualifying minority interest and regulatory adjustments. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de PortugalProvisionstoProvisions and impairments (net of reversals) divided by total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalLoanLossImpairments and value adjustments in customer loans divided by gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de PortugalRisk-weightsAverage risk-weight of exposure amount. Risk-weighted assets are all assets. Séries Longas para o Setor Bancário Português, Banco de PortugalLiquidity RatioCash and cash balance/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalLiquidity RatioCash and cash balance/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalLiquidity RatioCash and		Internal supervisory data. Banco de Portugal: Séries Longas para o
Debt Funding Cost rate (Debt) Annualized interest and other similar charges from debt securities issued and other financial liabilities divided by total liabilities represented by debt titles. Internal supervisory data, Banco de Portugal; Solvency Ratio Tier 1 Capital divided by total risk-weighted assets. Tier 1 capital is the sum of common shares and stock surplus, retained earnings, other comprehensive income, qualifying minority interest, capital instruments meeting the criteria from AT1 and related surplus, additional qualifying minority interest and regulatory adjustments. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Provisions to Provisions and impairments (net of reversals) divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Loan Loss Impairments and value adjustments in customer loans. Séries Longas para o Setor Bancário Português, Banco de Portugal Risk-weights Average risk-weight of exposure amount. Risk-weighted assets are all (RW) Cash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Liquidity Ratio Cash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Return on Equity Net profit or loss divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Size Logarithm of total assets. Séries Longa		Setor Bancário Português, Banco de Portugal
Cost rate (Debt) issued and other financial liabilities divided by total liabilities represented by debt titles. Internal supervisory data, Banco de Portugal; Séries Longas para o Setor Bancário Português, Banco de Portugal Solvency Ratio Tier 1 Capital divided by total risk-weighted assets. Tier 1 capital is the sum of common shares and stock surplus, retained earnings, other comprehensive income, qualifying minority interest, capital instruments meeting the criteria from AT1 and related surplus, additional qualifying minority interest and regulatory adjustments. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Provisions to Provisions and impairments (net of reversals) divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Loan Loss Impairments and value adjustments in customer loans divided by gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de Portugal Risk-weights Average risk-weight of exposure amount. Risk-weighted assets are all (RW) assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Liquidity Ratio Cash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Size Logarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Return on Equity Net profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de Portugal<	Debt Funding	Annualized interest and other similar charges from debt securities
 represented by debt titles. Internal supervisory data, Banco de Portugal; Séries Longas para o Setor Bancário Português, Banco de Portugal Solvency Ratio Tier I Capital divided by total risk-weighted assets. Tier 1 capital is the sum of common shares and stock surplus, retained earnings, other comprehensive income, qualifying minority interest, capital instruments meeting the criteria from AT1 and related surplus, additional qualifying minority interest and regulatory adjustments. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Provisions to Provisions and impairments (net of reversals) divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Loan Loss Impairments and value adjustments in customer loans divided by Reserves gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de Portugal Risk-weights Average risk-weight of exposure amount. Risk-weighted assets are all assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Liquidity Ratio Cash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Return on Equity Net profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de Portugal Size Logarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Colsing rate for the overnight maturity of the 1-day interbank interest rate for the Euro zone. Statistical Data Warehouse, ECB (EONIA) Dow Jones Dow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data War	Cost rate (Debt)	issued and other financial liabilities divided by total liabilities
Portugal; Séries Longas para o Setor Bancário Português, Banco de Portugal Solvency Ratio Tier 1 Capital divided by total risk-weighted assets. Tier 1 capital is the sum of common shares and stock surplus, retained earnings, other comprehensive income, qualifying minority interest, capital instruments meeting the criteria from AT1 and related surplus, additional qualifying minority interest and regulatory adjustments. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Provisions to Provisions and impairments (net of reversals) divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Loan Loss Impairments and value adjustments in customer loans divided by gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de Portugal Risk-weights Average risk-weight of exposure amount. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Liquidity Ratio Cash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Return on Equity Net profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de Portugal Size Logarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Cosing rate for the overnight maturity of the 1-day interbank interest rate for the Euro zone. Statistic	()	represented by debt titles. Internal supervisory data, Banco de
PortugalSolvency RatioTier 1 Capital divided by total risk-weighted assets. Tier 1 capital is the sum of common shares and stock surplus, retained earnings, other comprehensive income, qualifying minority interest, capital instruments meeting the criteria from AT1 and related surplus, additional qualifying minority interest and regulatory adjustments. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de PortugalProvisions AssetsProvisions and impairments (net of reversals) divided by total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalLoan LossImpairments and value adjustments in customer loans divided by gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de PortugalRisk-weights (RW)Average risk-weight of exposure amount. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de PortugalLiquidity RatioCash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalReturn on Equity (ROE)Net profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de PortugalSizeLogarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalSizeLogarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalDow JonesDow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical		Portugal; Séries Longas para o Setor Bancário Português, Banco de
Solvency Ratio Tier 1 Capital divided by total risk-weighted assets. Tier 1 capital is the sum of common shares and stock surplus, retained earnings, other comprehensive income, qualifying minority interest, capital instruments meeting the criteria from AT1 and related surplus, additional qualifying minority interest and regulatory adjustments. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Provisions to Provisions and impairments (net of reversals) divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Loan Loss Impairments and value adjustments in customer loans divided by gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de Portugal Risk-weights Average risk-weight of exposure amount. Risk-weighted assets are all assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Liquidity Ratio Cash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Return on Equity Net profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de Portugal Size Logarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Euro OverNight Index Average (EONIA) Dow Jones Dow Jones Dow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 f		Portugal
 is the sum of common shares and stock surplus, retained earnings, other comprehensive income, qualifying minority interest, capital instruments meeting the criteria from AT1 and related surplus, additional qualifying minority interest and regulatory adjustments. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Provisions to Provisions and impairments (net of reversals) divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Loan Loss Impairments and value adjustments in customer loans divided by gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de Portugal Risk-weights Average risk-weight of exposure amount. Risk-weighted assets are all assets held by the bank weighted by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Liquidity Ratio Cash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Return on Equity Net profit or loss divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Return on Equity Net profit or loss divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Size Logarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Size Logarithm of total assets. Séries Longas para o Setor Bancário Português, Bancó de Portugal Size Logarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Size Logarithm of total assets. Séries Longas para o Setor Bancário Português, Bancó de Portugal Size Logarithm of total assets. Séries Longas para o Setor Bancário Português, Bancó de Portugal Dow Jones Dow Jones Euro Stoxx Price Inde	Solvency Ratio	Tier 1 Capital divided by total risk-weighted assets. Tier 1 capital
other comprehensive income, qualifying minority interest, capital instruments meeting the criteria from AT1 and related surplus, additional qualifying minority interest and regulatory adjustments. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Provisions to Provisions to Assets Séries Longas para o Setor Bancário Português, Banco de Portugal Loan Loss Impairments and value adjustments in customer loans divided by gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de Portugal Risk-weights Average risk-weight of exposure amount. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Liquidity Ratio Cash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Return on Equity Net profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de Portugal Size Logarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Euro OverNight rate for the Euro zone. Statistical Data Warehouse, ECB Dow Jones Dow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Wareh	2	is the sum of common shares and stock surplus, retained earnings,
 instruments meeting the criteria from AT1 and related surplus, additional qualifying minority interest and regulatory adjustments. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Provisions to Provisions and impairments (net of reversals) divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Loan Loss Impairments and value adjustments in customer loans divided by Reserves gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de Portugal Risk-weights Average risk-weight of exposure amount. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Liquidity Ratio Cash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Return on Equity Net profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de Portugal Size Logarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Size Logarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Euro OverNight Closing rate for the overnight maturity of the 1-day interbank interest rate for the Euro zone. Statistical Data Warehouse, ECB V2TX Euro Stocc Volatility (VSTOXX) index computed by QON-TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse Group Deposits Annual- Observed annualised agreed rates for new deposits (TAA, taxa acordada anualizada in its Porturusee denomination) "Estatísticas 		other comprehensive income, qualifying minority interest, capital
additional qualifying minority interest and regulatory adjustments. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de PortugalProvisionstoProvisions and impairments (net of reversals) divided by total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalLoanLossImpairments and value adjustments in customer loans divided by gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de PortugalRisk-weightsAverage risk-weight of exposure amount. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de PortugalLiquidity RatioCash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalReturn on EquityNet profit or loss divided by total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalSizeLogarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalSizeLogarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalEuroOverNight Index Average (EONIA)Dow JonesDow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECBV2TXEuro Stocc Volatility (VSTOXX) index computed by QON- TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse GroupDeposits Annual- ized Arreed Rate acordada anualized aire is		instruments meeting the criteria from AT1 and related surplus,
Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de PortugalProvisionstoProvisionstoAssetsassets. Séries Longas para o Setor Bancário Português, Banco de PortugalLoanLossImpairments and value adjustments in customer loans divided by gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de PortugalRisk-weightsAverage risk-weight of exposure amount. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de PortugalLiquidity RatioCash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalReturn on EquityNet profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de PortugalSizeLogarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalEuroOverNight Closing rate for the overnight maturity of the 1-day interbank interest rate for the Euro zone. Statistical Data Warehouse, ECBV2TXEuro Stocc Volatility (VSTOXX) index computed by QON- TIGO. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECBV2TXEuro Stocc Volatility (VSTOXX) index computed by QON- TIGO. Divided annualized an rulaized agreed rates for new deposits (TAA, taxa acordada anualizada in its Portugues denomination) "Estatícticas		additional qualifying minority interest and regulatory adjustments.
respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Provisions to Provisions and impairments (net of reversals) divided by total Assets assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Loan Loss Impairments and value adjustments in customer Ioans divided by Reserves gross carrying amount of customer Ioans. Séries Longas para o Setor Bancário Português, Banco de Portugal Risk-weights Average risk-weight of exposure amount. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Liquidity Ratio Cash and cash balances/Ioans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Return on Equity Net profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de Portugal Size Logarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Euro OverNight Closing rate for the overnight maturity of the 1-day interbank interest rate for the Euro zone. Statistical Data Warehouse, ECB (EONIA) Dow Jones Dow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECB V2TX Euro Stocc Volatility (VSTOXX) index computed by QON- TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse Group Deposits Annual- Observed annualised agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination) "Estatícticas		Risk-weighted assets are all assets held by the bank weighted by the
Banco de PortugalProvisionstoAssetsProvisions and impairments (net of reversals) divided by total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalLoanLossImpairments and value adjustments in customer loans divided by gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de PortugalRisk-weightsAverage risk-weight of exposure amount. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de PortugalLiquidity RatioCash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalReturn on Equity (ROE)Net profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de PortugalSizeLogarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalEuro OverNight Index Average (EONIA)Closing rate for the overnight maturity of the 1-day interbank interest rate for the Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECBV2TXEuro Stocc Volatility (VSTOXX) index computed by QON- TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deeutsche Börse GroupDeposits Annual- ized Agreed RateObserved annualised agreed rates for new deposits (TAA, taxa acordada anualized ain its Portuguese denomination)		respective credit risk. Séries Longas para o Setor Bancário Português,
ProvisionstoProvisions and impairments (net of reversals) divided by total assetsAssetsassets. Séries Longas para o Setor Bancário Português, Banco de PortugalLoanLossImpairments and value adjustments in customer loans divided by gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de PortugalRisk-weightsAverage risk-weight of exposure amount. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de PortugalLiquidity RatioCash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalReturn on Equity (ROE)Net profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de PortugalSizeLogarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalEuroOverNight Index Average (EONIA)Dow JonesDow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECBV2TXEuro Stocc Volatility (VSTOXX) index computed by QON- TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse GroupDeposits Annual- ized Agreed RateObserved annualised agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination)		Banco de Portugal
Assets assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Loan Loss Impairments and value adjustments in customer loans divided by gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de Portugal Risk-weights Average risk-weight of exposure amount. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Liquidity Ratio Cash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Return on Equity Net profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de Portugal Size Logarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Euro OverNight Index Average rate for the Euro zone. Statistical Data Warehouse, ECB (EONIA) Dow Jones Dow Jones Dow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECB V2TX Euro Stocc Volatility (VSTOXX) index computed by QON- TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse Group Deposits Annual- ized Agreed Rate Observed annualised agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denominatio	Provisions to	Provisions and impairments (net of reversals) divided by total
PortugalLoanLossImpairments and value adjustments in customer loans divided by gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de PortugalRisk-weightsAverage risk-weight of exposure amount. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de PortugalLiquidity RatioCash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalReturn on EquityNet profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de PortugalSizeLogarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalEuro OverNight Index Average (EONIA)Closing rate for the overnight maturity of the 1-day interbank interest rate for the Euro zone. Statistical Data Warehouse, ECBV2TXEuro Stocc Volatility (VSTOXX) index computed by QON- TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse GroupDeposits Annual- ized Agreed RateObserved annualised agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination) "Estatísticas	Assets	assets. Séries Longas para o Setor Bancário Português, Banco de
LoanLossImpairments and value adjustments in customer loans divided by gross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de PortugalRisk-weightsAverage risk-weight of exposure amount. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de PortugalLiquidity RatioCash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalReturn on EquityNet profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de PortugalSizeLogarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalEuroOverNight Index Average (EONIA)Dow JonesDow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECBV2TXEuro Stocc Volatility (VSTOXX) index computed by QON- TIGO. Divided Biy 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse GroupDeposits Annual- ized Agreed RateObserved annualized agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination) "Estatística		Portugal
Reservesgross carrying amount of customer loans. Séries Longas para o Setor Bancário Português, Banco de PortugalRisk-weightsAverage risk-weight of exposure amount. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de PortugalLiquidity RatioCash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalReturn on Equity (ROE)Net profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de PortugalSizeLogarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalEuro OverNight Index Average (EONIA)Closing rate for the overnight maturity of the 1-day interbank interest rate for the Euro zone. Statistical Data Warehouse, ECB (EONIA)Dow JonesDow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECBV2TXEuro Stocc Volatility (VSTOXX) index computed by QON- TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse GroupDeposits Annual- ized Agreed Rate acordada anualizada in its Portuguese denomination)"Estatísticas	Loan Loss	Impairments and value adjustments in customer loans divided by
Bancário Português, Banco de PortugalRisk-weightsAverage risk-weight of exposure amount. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de PortugalLiquidity RatioCash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalReturn on EquityNet profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de PortugalSizeLogarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalEuro OverNight Index Average (EONIA)Closing rate for the overnight maturity of the 1-day interbank interest rate for the Euro zone. Statistical Data Warehouse, ECB (EONIA)Dow JonesDow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECBV2TXEuro Stocc Volatility (VSTOXX) index computed by QON- TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse GroupDeposits Annual- ized Agreed Rate acordada anualizada in its Portuguese denomination)"Estatísticas	Reserves	gross carrying amount of customer loans. Séries Longas para o Setor
Risk-weights (RW)Average risk-weight of exposure amount. Risk-weighted assets are all assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de PortugalLiquidity RatioCash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalReturn on EquityNet profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de PortugalSizeLogarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalEuroOverNight Index Average (EONIA)Dow JonesDow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECBV2TXEuro Stocc Volatility (VSTOXX) index computed by QON- TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse GroupDeposits Annual- ized Agreed RateObserved annualised agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination)		Bancário Português, Banco de Portugal
(RW) assets held by the bank weighted by the respective credit risk. Séries Longas para o Setor Bancário Português, Banco de Portugal Liquidity Ratio Cash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Return on Equity Net profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de Portugal Size Logarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Euro OverNight Closing rate for the overnight maturity of the 1-day interbank interest rate for the Euro zone. Statistical Data Warehouse, ECB (EONIA) Dow Jones Dow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECB V2TX Euro Stocc Volatility (VSTOXX) index computed by QON-TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse Group Deposits Annual- ized Agreed Rate Observed annualised agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination) "Estatísticas	Risk-weights	Average risk-weight of exposure amount. Risk-weighted assets are all
Longas para o Setor Bancário Português, Banco de PortugalLiquidity RatioCash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalReturn on EquityNet profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de PortugalSizeLogarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalEuroOverNightClosing rate for the overnight maturity of the 1-day interbank interest rate for the Euro zone. Statistical Data Warehouse, ECB (EONIA)Dow JonesDow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECBV2TXEuro Stocc Volatility (VSTOXX) index computed by QON- TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse GroupDeposits Annual- ized Agreed RateObserved annualised agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination) "Estatísticas	(RW)	assets held by the bank weighted by the respective credit risk. Séries
Liquidity RatioCash and cash balances/loans to central banks divided by total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalReturn on EquityNet profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de PortugalSizeLogarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de PortugalEuroOverNightIndexAverage rate for the Euro zone. Statistical Data Warehouse, ECB (EONIA)Dow JonesDow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECBV2TXEuro Stocc Volatility (VSTOXX) index computed by QON- TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse GroupDeposits Annual- ized Agreed RateObserved annualised agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination)		Longas para o Setor Bancário Português, Banco de Portugal
assets. Series Longas para o Setor Bancário Português, Banco de Portugal Return on Equity Net profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de Portugal Size Logarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Euro OverNight Index Average (EONIA) Dow Jones Dow Jones Dow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECB V2TX Euro Stocc Volatility (VSTOXX) index computed by QON- TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse Group Deposits Annual- ized Agreed Rate Observed annualised agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination) "Estatísticas	Liquidity Ratio	Cash and cash balances/loans to central banks divided by total
Portugal Return on Equity Net profit or loss divided by total equity. Séries Longas para o Setor Bancário Português, Banco de Portugal Size Logarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Euro OverNight Closing rate for the overnight maturity of the 1-day interbank interest rate for the Euro zone. Statistical Data Warehouse, ECB Dow Jones Dow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECB V2TX Euro Stocc Volatility (VSTOXX) index computed by QON- TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse Group Deposits Annual- ized Agreed Rate Observed annualised agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination) "Estatísticas		assets. Séries Longas para o Setor Bancário Português, Banco de
Return on Equity Net profit or loss divided by total equity. Series Longas para o Setor (ROE) Bancário Português, Banco de Portugal Size Logarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Euro OverNight Index Average (EONIA) Closing rate for the overnight maturity of the 1-day interbank interest rate for the Euro zone. Statistical Data Warehouse, ECB Dow Jones Dow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECB V2TX Euro Stocc Volatility (VSTOXX) index computed by QON- TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse Group Deposits Annual- ized Agreed Rate Observed annualised agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination) "Estatísticas		Portugal
(ROE) Bancario Portugues, Banco de Portugal Size Logarithm of total assets. Séries Longas para o Setor Bancário Português, Banco de Portugal Euro OverNight Closing rate for the overnight maturity of the 1-day interbank interest rate for the Euro zone. Statistical Data Warehouse, ECB (EONIA) Dow Jones Dow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECB V2TX Euro Stocc Volatility (VSTOXX) index computed by QON- TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse Group Deposits Annual- ized Agreed Rate Observed annualised agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination) "Estatísticas	Return on Equity	Net profit or loss divided by total equity. Series Longas para o Setor
Size Logarithm of total assets. Series Longas para o Setor Bancario Português, Banco de Portugal Euro OverNight Closing rate for the overnight maturity of the 1-day interbank interest rate for the Euro zone. Statistical Data Warehouse, ECB (EONIA) Dow Jones Dow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECB V2TX Euro Stocc Volatility (VSTOXX) index computed by QON- TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse Group Deposits Annual- ized Agreed Rate Observed annualised agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination) "Estatísticas	(ROE)	Bancario Portugues, Banco de Portugal
Fortugues, Banco de Portugal Euro OverNight Index Average rate for the Euro zone. Statistical Data Warehouse, ECB (EONIA) Dow Jones Uters Dow Jones Dow Jones Dow Jones	Size	Logarithm of total assets. Series Longas para o Setor Bancário
Euro OverNight Closing rate for the overnight maturity of the 1-day interbank interest Index Average rate for the Euro zone. Statistical Data Warehouse, ECB (EONIA) Dow Jones Dow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECB V2TX Euro Stocc Volatility (VSTOXX) index computed by QON-TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse Group Deposits Annual- Observed annualised agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination) "Estatísticas		Portugues, Banco de Portugal
Index Average rate for the Euro zone. Statistical Data Warehouse, ECB (EONIA) Dow Jones Dow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECB V2TX Euro Stocc Volatility (VSTOXX) index computed by QON-TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse Group Deposits Annual-ized Agreed Rate Observed annualized agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination) "Estatísticas	Euro OverNight	Closing rate for the overnight maturity of the 1-day interbank interest
(EONIA) Dow Jones Dow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECB V2TX Euro Stocc Volatility (VSTOXX) index computed by QON-TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse Group Deposits Annual-ized Agreed Rate Observed annualized agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination) "Estatísticas	Index Average	rate for the Euro zone. Statistical Data Warehouse, ECB
Dow Jones Dow Jones Euro Stoxx Price Index. Historical close, average of observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECB V2TX Euro Stocc Volatility (VSTOXX) index computed by QON-TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse Group Deposits Annual-ized Agreed Rate Observed annualised agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination) "Estatísticas		Deve lance Free Charles Index Illisterial alore success
or observations through period. Divided by 100 for easier interpretation. Statistical Data Warehouse, ECB V2TX Euro Stocc Volatility (VSTOXX) index computed by QON-TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse Group Deposits Annual-ized Agreed Rate Observed annualised agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination) "Estatísticas	Dom Joues	Dow Jones Euro Stoxx Price Index. Historical close, average
V2TX Euro Stocc Volatility (VSTOXX) index computed by QON-TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deposits Annual-Observed annualised agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination)		of observations through period. Divided by 100 for easier
V21X Luto Stoce volatinty (V310XX) index computed by QON- TIGO. Divided by 100 for easier interpretation. STOXX, QONTIGO, Deutsche Börse Group Deposits Annual- ized Agreed Rate Observed annualised agreed rates for new deposits (TAA, taxa acordada anualizada in its Portuguese denomination) "Estatísticas		Fure Stoce Velatility (VSTOXX) index computed by OON
Deposits Annual- ized Agreed Rate acordada anualizada in its Portuguese denomination) "Estatísticas	VZIA	TICO Divided by 100 for easier interpretation STOXY CONTICO
Deposits Annual- ized Agreed Rate acordada anualizada in its Portuguese denomination) "Estatísticas		Deutsche Börse Group
ized Agreed Rate acordada anualizada in its Portuguese denomination) "Estatísticas	Deposits Annual	Observed annualised agreed rates for new denosits (TAA taxa
TEED METEED MALE ACTIVITIES AND A THE TEACHTER AND A THE ACTIVITIES AND A THE ACTIVITES AND A THE ACTIVITIES AND A	ized Agreed Rate	acordada anualizada in its Portuguese denomination) "Estatísticas
de balanco e de taxas de juro das instituições financeiras monetárias"	ized Agreed Male	de balanco e de taxas de juro das instituições financeiras monetárias"
Banco de Portugal		Banco de Portugal

27

References

- Aldasoro, Iñaki, Chun Hee Cho, and Kyounghoon Park (2022). "Bank solvency risk and funding cost interactions: Evidence from Korea." *Journal of Banking & Finance*, 134, 106348.
- Andrews, Isaiah, James H. Stock, and Liyang Sun (2019). "Weak Instruments in Instrumental Variables Regression: Theory and Practice." *Annual Review of Economics*, 11.
- Arnould, Guillaume, Giuseppe Avignone, Cosimo Pancaro, and Dawid Żochowski (2021). "Bank funding costs and solvency." *The European Journal of Finance*, 0, 1–33.
- Aymanns, Christoph, Carlos Caceres, Christina Daniel, and MissLiliana Schumacher (2016). *Bank Solvency and Funding Cost*. International Monetary Fund.
- Babihuga, Rita, Marco Spaltro, and Krishna Srinivasan (2014). "Bank Funding Costs for International Banks." *IMF Working Papers*, 2014(071).
- Bahaj, Saleem and Frederic Malherbe (2020). "The Forced Safety Effect: How Higher Capital Requirements Can Increase Bank Lending." *The Journal of Finance*, 75(6).
- Boot, Arnould W. A., Stuart I. Greenbaum, and Anjan V. Thakor (1993).
 "Reputation and Discretion in Financial Contracting." *The American Economic Review*, 83(5), 1165–1183.
- Chen, Jie and Arjun K. Gupta (2012). *Parametric Statistical Change Point Analysis*. Birkhäuser Boston, Boston.
- Couaillier, Cyril, Marco Lo Duca, Alessio Reghezza, Costanza Rodriguez d'Acri, and Alessandro Scopelliti (2021). "Bank capital buffers and lending in the euro area during the pandemic." *Financial Stability Review*, 2.
- Couchinho, Ana, Nuno Ribeiro, Paulo Esteves, Bruno Nascimento, Catarina Ramos, Luís Rodrigues, and Ricardo Torre (2019). "Séries Longas – Setor Bancário Português 1990-2018." *Banco de Portugal*, p. 129.
- Davies, Robert B. (1987). "Hypothesis Testing when a Nuisance Parameter is Present Only Under the Alternatives." Biometrika, 74(1).
- Demirgüç-Kunt, Asli and Harry Huizinga (2004). "Market discipline and deposit insurance." *Journal of Monetary Economics*, 51(2), 375–399.
- Elyasiani, Elyas and Jason Keegan (2017). "Market Discipline in the Secondary Bond Market: The Case of Systemically Important Banks."
- Gambacorta, Leonardo and Hyun Song Shin (2016). "Why bank capital matters for monetary policy." *BIS Working Papers*, (558).
- Hasan, Iftekhar, Liuling Liu, and Gaiyan Zhang (2016). "The Determinants of Global Bank Credit-Default-Swap Spreads." *Journal of Financial Services Research*, 50(3).
- King, Thomas B. (2008). "Discipline and Liquidity in the Interbank Market." *Journal of Money, Credit and Banking*, 40(2-3), 295–317.
- Muggeo, Vito (2008). "Segmented: An R Package to Fit Regression Models With Broken-Line Relationships." *R News*, 8.

- 29 The solvency and funding cost nexus the role of market stigma for buffer usability
- Muggeo, Vito M. R. (2003). "Estimating regression models with unknown breakpoints." *Statistics in Medicine*, 22(19), 3055–3071.
- Muggeo, Vito M. R. (2016). "Testing with a nuisance parameter present only under the alternative: a score-based approach with application to segmented modelling." *Journal of Statistical Computation and Simulation*, 86(15), 3059–3067.
- Muggeo, Vito M.R. (2017). "Interval estimation for the breakpoint in segmented regression: a smoothed score-based approach." *Australian & New Zealand Journal of Statistics*, 59(3).
- on Banking Supervision, Basel Committee (ed.) (2015). *Making supervisory stress tests more macroprudential: considering liquidity and solvency interactions and systemic risk*. No. 29 in Working Paper / Basel Committee on Banking Supervision, Bank for International Settlements, Bank for International Settlements, Basel.
- Oura, Hiroko, Brenda González-Hermosillo, Jorge A. Chan-Lau, Tryggvi Gudmundsson, and Nico Valckx (2013). "Changes in Bank Funding Patterns and Financial Stability Risks." Rochester, NY.
- Rua, António (2017). "Dating the Portuguese business cycle." *Economic Bulletin* and Financial Stability Report Articles and Banco de Portugal Economic Studies, p. 16.
- Scheines, Richard, Gregory F. Cooper, Changwon Yoo, and Tianjiao Chu (2001). "Piecewise Linear Instrumental Variable Estimation of Causal Influence." In International Workshop on Artificial Intelligence and Statistics, pp. 265–271, PMLR. ISSN: 2640-3498.
- Schmitz, Stefan, Michael Sigmund, and Laura Valderrama (2017). "Bank Solvency and Funding Cost: New Data and New Results."
- Stock, James H. and Motohiro Yogo (2005). "Testing for Weak Instruments in Linear IV Regression." In Identification and Inference for Econometric Models: Essays in Honor of Thomas Rothenberg, edited by Donald W. K. Andrews and James H. Stock, pp. 80–108. Cambridge University Press, Cambridge.
- Wu, Jing Cynthia and Fan Dora Xia (2020). "Negative interest rate policy and the yield curve." Journal of Applied Econometrics, 35(6), 653–672.

Working Papers

2020

- 1|20 On-site inspecting zombie lending Diana Bonfim | Geraldo Cerqueiro | Hans Degryse | Steven Ongena
- 2 2 Labor earnings dynamics in a developing economy with a large informal sector
 Diego B. P. Gomes | Felipe S. lachan | Cezar Santos
- 3|20 Endogenous growth and monetary policy: how do interest-rate feedback rules shape nominal and real transitional dynamics? Pedro Mazeda Gil | Gustavo Iglésias
- 4|20 Types of International Traders and the Network of Capital Participations João Amador | Sónia Cabral | Birgitte Ringstad
- 5|20 Forecasting tourism with targeted predictors in a data-rich environment Nuno Lourenço | Carlos Melo Gouveia | António Rua
- 6|20 The expected time to cross a threshold and its determinants: A simple and flexible framework

Gabriel Zsurkis | João Nicolau | Paulo M. M. Rodrigues

- 7|20 A non-hierarchical dynamic factor model for three-way data Francisco Dias | Maximiano Pinheiro | António Rua
- 8|20 Measuring wage inequality under right censoring João Nicolau | Pedro Raposo | Paulo M. M. Rodrigues
- 9|20 Intergenerational wealth inequality: the role of demographics António Antunes | Valerio Ercolani
- 10|20 Banks' complexity and risk: agency problems and diversification benefits Diana Bonfim | Sónia Felix

- 11|20 The importance of deposit insurance credibility Diana Bonfim | João A. C. Santos
- 12|20 Dream jobs Giordano Mion | Luca David Opromolla | Gianmarco I.P. Ottaviano
- 13|20 The DEI: tracking economic activity daily during the lockdown Nuno Lourenço | António Rua
- 14|20 An economic model of the Covid-19 pandemic with young and old agents: Behavior, testing and policies Luiz Brotherhood | Philipp Kircher | Cezar Santos | Michèle Tertilt
- 15|20 Slums and Pandemics Luiz Brotherhood | Tiago Cavalcanti | Daniel Da Mata | Cezar Santos
- 16|20 Assessing the Scoreboard of the EU Macroeconomic Imbalances Procedure: (Machine) Learning from Decisions
 Tiago Alves | João Amador | Francisco Gonçalves
- 17|20 Climate Change Mitigation Policies: Aggregate and Distributional Effects Tiago Cavalcanti | Zeina Hasna | Cezar Santos
- 18|20 Heterogeneous response of consumers to income shocks throughout a financial assistance program Nuno Alves | Fátima Cardoso | Manuel Coutinho Pereira
- **19|20** To change or not to change: the impact of the law on mortgage origination Ana Isabel Sá

2021

- 1|21 Optimal Social Insurance: Insights from a Continuous-Time Stochastic Setup João Amador | Pedro G. Rodrigues
- 2|21 Multivariate Fractional Integration Tests allowing for Conditional Heteroskedasticity withan Application to Return Volatility and Trading

Marina Balboa | Paulo M. M. Rodrigues Antonio Rubia | A. M. Robert Taylor

3 21 The Role of Macroprudential Policy in Times of Trouble

Jagjit S. Chadha | Germana Corrado | Luisa Corrado | Ivan De Lorenzo Buratta

4|21 Extensions to IVX Methodsnof Inference for Return Predictability Matei Demetrescu | Iliyan Georgiev | Paulo

M. M. Rodrigues | A.M. Robert Taylor

- 5|21 Spectral decomposition of the information about latent variables in dynamic macroeconomic models Nikolay Iskrev
- 6|21 Institutional Arrangements and Inflation Bias: A Dynamic Heterogeneous Panel Approach

Vasco Gabriel | Ioannis Lazopoulos | Diana Lima

- 7|21 Assessment of the effectiveness of the macroprudential measures implemented in the context of the Covid-19 pandemic Lucas Avezum | Vítor Oliveiral | Diogo Serra
- 8|21 Risk shocks, due loans, and policy options: When less is more! Paulo Júlio | José R. Maria | Sílvia Santos
- 9|21 Sovereign-Bank Diabolic Loop: The Government Procurement Channel! Diana Bonfim | Miguel A. Ferreira | Francisco Queiró | Sujiao Zhao

- 10|21 Assessing the effectiveness of the Portuguese borrower-based measure in the Covid-19 context Katja Neugebauer | Vítor Oliveira | Ângelo Ramos
- 11|21 Scrapping, Renewable Technology Adoption, and Growth Bernardino Adão | Borghan Narajabad | Ted Temzelides
- 12|21 The Persistence of Wages Anabela Carneiro | Pedro Portugal | Pedro Raposo | Paulo M.M. Rodrigues
- 13|21 Serial Entrepreneurs, the Macroeconomy and top income inequality Sónia Félix | Sudipto Karmakar | Petr Sedláček
- 14|21 COVID-19, Lockdowns and International Trade: Evidence from Firm-Level Data João Amador | Carlos Melo Gouveia | Ana Catarina Pimenta
- 15|21 The sensitivity of SME's investment and employment to the cost of debt financing Diana Bonfim | Cláudia Custódio | Clara Raposo
- 16|21 The impact of a macroprudential borrower based measure on households' leverage and housing choices Daniel Abreu | Sónia Félix | Vítor Oliveira | Fátima Silva
- 17|21 Permanent and temporary monetary policy shocks and the dynamics of exchange rates Alexandre Carvalho | João Valle e Azevedo | Pedro Pires Ribeiro
- 18|21 On the Cleansing Effect of Recessions and Government Policy: Evidence from Covid-19

Nicholas Kozeniauskas | Pedro Moreira | Cezar Santos

- 19|21 Trade, Misallocation, and Capital Market Integration Laszlo Tetenyi
- 20|21 Not All Shocks Are Created Equal: Assessing Heterogeneity in the Bank Lending Channel Laura Blattner | Luísa Farinha | Gil Nogueira

2022

- 1|22 Business cycle clocks: Time to get circular Nuno Lourenço | António Rua
- 2 22 The Augmented Bank Balance-Sheet Channel of Monetary Policy
 Christian Bittner | Diana Bonfim | Florian Heider | Farzad Saidi | Glenn Schepens | Carla Soares
- 3|22 Optimal cooperative taxation in the global economy V. V. Chari | Juan Pablo Nicolini | Pedro Teles
- 4|22 How Bad Can Financial Crises Be? A GDP Tail Risk Assessment for Portugal Ivan De Lorenzo Buratta | Marina Feliciano | Duarte Maia
- 5|22 Comparing estimated structural models of different complexities: What do we learn? Paulo Júlio | José R. Maria
- 6|22 Survival of the fittest: Tourism Exposure and Firm Survival

Filipe B. Caires | Hugo Reis | Paulo M. M. Rodrigues

- 21|21 Coworker Networks and the Labor Market Outcomes of Displaced Workers: Evidence from Portugal Jose Garcia-Louzao | Marta Silva
- 22|21 Markups and Financial Shocks Philipp Meinen | Ana Cristina Soares
- 7|22 Mind the Build-up: Quantifying Tail Risks for Credit Growth in Portugal
 Ivan de Lorenzo Buratta | Marina Feliciano | Duarte Maia
- 8 22 Forgetting Approaches to Improve Forecasting Robert Hill | Paulo M. M. Rodrigues
- 9|22 Determinants of Cost of Equity for listed euro area banks Gabriel Zsurkis
- 10|22 Real effects of imperfect bank-firm matching Luísa Farinha | Sotirios Kokas | Enrico Sette | Serafeim Tsoukas
- 11|22 The solvency and funding cost nexus the role of market stigma for buffer usability Helena Carvalho | Lucas Avezum | Fátima Silva